

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Belke, Ansgar; Domnick, Clemens; Gros, Daniel

# Working Paper Business cycle synchronization in the EMU: Core vs. periphery

Ruhr Economic Papers, No. 659

**Provided in Cooperation with:** RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

*Suggested Citation:* Belke, Ansgar; Domnick, Clemens; Gros, Daniel (2016) : Business cycle synchronization in the EMU: Core vs. periphery, Ruhr Economic Papers, No. 659, ISBN 978-3-86788-765-6, RWI - Leibniz-Institut für Wirtschaftsforschung, Essen, https://doi.org/10.4419/86788765

This Version is available at: https://hdl.handle.net/10419/148387

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# RUHR ECONOMIC PAPERS

Ansgar Belke Clemens Domnick Daniel Gros



# Imprint

#### Ruhr Economic Papers

Published by

Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany RWI Leibniz-Institut für Wirtschaftsforschung Hohenzollernstr. 1-3, 45128 Essen, Germany

#### Editors

Prof. Dr. Thomas K. Bauer
RUB, Department of Economics, Empirical Economics
Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de
Prof. Dr. Wolfgang Leininger
Technische Universität Dortmund, Department of Economic and Social Sciences
Economics - Microeconomics
Phone: +49 (0) 231/7 55-3297, e-mail: W.Leininger@tu-dortmund.de
Prof. Dr. Volker Clausen
University of Duisburg-Essen, Department of Economics
International Economics
Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de
Prof. Dr. Roland Döhrn, Prof. Dr. Manuel Frondel, Prof. Dr. Jochen Kluve
RWI, Phone: +49 (0) 201/81 49-213, e-mail: presse@rwi-essen.de

#### Editorial Office

Sabine Weiler

RWI, Phone: +49 (0) 201/81 49-213, e-mail: sabine.weiler@rwi-essen.de

#### Ruhr Economic Papers #659

Responsible Editor: Volker Clausen

All rights reserved. Bochum, Dortmund, Duisburg, Essen, Germany, 2016

ISSN 1864-4872 (online) - ISBN 978-3-86788-765-6

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

# Ruhr Economic Papers #659

Ansgar Belke, Clemens Domnick, and Daniel Gros

# Business Cycle Synchronization in the EMU: Core vs. Periphery



# Bibliografische Informationen der Deutschen Nationalbibliothek

Die Deutsche Bibliothek verzeichnet diese Publikation in der deutschen Nationalbibliografie; detaillierte bibliografische Daten sind im Internet über: http://dnb.d-nb.de abrufbar.

http://dx.doi.org/10.4419/86788765 ISSN 1864-4872 (online) ISBN 978-3-86788-765-6 Ansgar Belke, Clemens Domnick, and Daniel Gros<sup>1</sup>

# **Business Cycle Synchronization in the EMU:** Core vs. Periphery

# Abstract

This paper examines business cycle synchronization in the European Monetary Union with a special focus on the core-periphery pattern in the aftermath of the crisis. Using a quarterly index for business cycle synchronization by Cerqueira (2013), our panel data estimates suggest that it is countries belonging to the core that are faced with increased synchronization among themselves after 2007Q4, whereas peripheral countries decreased synchronization with regards to the core, non-EMU countries and among themselves. Correlation coefficients and nonparametric local polynomial regressions corroborate these findings. The usual focus on co-movements and correlations might be misleading, however, since we also find large differences in the amplitude of national cycles. A strong common cycle can thus lead to large differences in cyclical positions even if national cycles are strongly correlated.

JEL Classification: E32, F15, R23

*Keywords: Business cycles; core-periphery; EMU; local polynomial regressions; synchronicity; common monetary policy* 

October 2016

<sup>1</sup> Ansgar Belke, UDE and Institute for the Study of Labor (IZA), Bonn; Clemens Domnick, UDE; Daniel Gros, Centre for European Policy Studies, Brussels. – The authors would like to thank Thomas Osowski, Pedro Cerqueira, Daniel Wissmann and participants of the Jean Monnet conference Economic Prospects for the European Union – Challenges for Economic Policy until the End of the Decade in Düsseldorf for helpful comments and suggestions. All remaining errors are our own. – All Correspondence to: Clemens Domnick, University of Duisburg-Essen, Universitätstraße 12, 45117 Essen, Germany, e-mail: clemens.domnick@gmail.com

## 1. Introduction

"Uniform monetary policy and inflexible exchange rates will create conflicts whenever cyclical conditions differ among the member countries." Martin Feldstein (1997)

The crisis in the European Monetary Union (EMU) brought a sharp economic division between core and peripheral members states to the limelight (European Commission, 2014). Whereas core countries such as Germany had experienced only temporary losses in output, peripheral countries such as Greece or Portugal continue to fight with weak economic activity and a stark rise in unemployment. Analyzing and assessing cross-country heterogeneity among EMU members play a key role for establishing a sustainable governance for the euro area. Hence, the core-periphery paradigma has been subject of much academic work (Blanchard *et al.*, 2015; Wortmann and Stahl, 2016; Cesaroni and De Santis, 2016).

The core-periphery distinction is especially relevant for monetary policy. Diverging economic trajectories pose a significant threat to the stability of the EMU, since the European Central Bank (ECB) can only react with a "one-size-fits-all" interest rate policy to stabilize output movements. The importance of synchronized business cycles for a well-functioning currency union is stressed in the theory of the Optimum Currency Area (OCA), pioneered by the work of Mundell (1961) and McKinnon (1963).<sup>1</sup> Firstly, the more aligned the business cycles of members of a monetary union are, the easier it is for the central bank to conduct stabilization policies (Clarida *et al.*, 1999; Rogoff, 1985). Secondly, a high degree of synchronization between national cycles may reduce the probability of asymmetric shock propagation across EMU members (Altavilla, 2004).

In a string of papers, Eichengreen (1990, 1991, 1993) concludes that the cost of giving up autonomy over monetary policy is especially high if business cycles of member states are only weakly correlated and alternative adjustment mechanisms, such as factor mobility, are not sufficiently available. To put it differently: the higher the synchronization of business cycles among member states is, the lower

<sup>&</sup>lt;sup>1</sup>See Mongelli (2008) for a theory on the evolution of the OCA idea.

is the cost of abandoning national monetary policy. As a result, business cycle synchronization is considered to be the most important OCA criterion (Gächter  $et \ al., 2012$ ).

Based on this reasoning, our paper adds to the vast literature on business cycle synchronization by providing a thorough analysis of how output co-moves among EMU member states. Specifically, we complement the literature by focusing on the relationship between two clusters within the EMU: a group of core countries, namely Germany, France, Austria, Finland and the Netherlands, and a group of peripheral countries, i.e. Portugal, Italy, Ireland, Greece and Spain.<sup>2</sup> Several studies pointed to a division of core and peripheral countries even before the introduction of the Euro (Bayoumi and Eichengreen, 1992; Dickerson *et al.*, 1998).<sup>3</sup> Lehwald (2013) concludes that business cycles for core and peripheral countries diverged after the introduction of the EMU, in line with Papageorgiou *et al.* (2010). Our main research questions are: what are the correlation patterns between the core and the periphery in the EMU, and how has the co-movement of output evolved over time, especially in the crisis period starting in 2007?

The bulk of earlier studies that examined business cycle synchronization relies either on calculating a correlation measure over the entire time period or over nonoverlapping subperiods of time (Furceri and Karras, 2008; Gouveia and Correia, 2008). The observed time window is often set arbitrarily and the correlation coefficients are prone to potential outliers biasing the results. We are able to overcome these problems by using a correlation index pioneered by Cerqueira and Martins (2009) that provides us with a correlation measure on a quarterly basis. In our empirical analysis, we start with a comparison of simple correlation coefficients, but continue with both parametric and nonparametric estimations in the spirit of

<sup>&</sup>lt;sup>2</sup>There exists no exact definition as to which countries belong to the core or to the periphery. With regards to the latter, the literature nearly unanimously includes Portugal, Spain, Ireland and Greece, with Italy being sometimes excluded (Lehwald, 2013). For the former, we follow a geographical definition and include France in the group of core countries, as opposed to studies clustering along country-specific economic policies (Schäfer, 2016).

<sup>&</sup>lt;sup>3</sup>The distinction between a core and a peripheral set of countries was also found for the other criteria suggested by the OCA theory, see Artis (2003). As Artis and Zhang (2002) point out, there exists a group of Southern peripheral countries, namely Spain, Italy, Portugal, and Greece, for which participating in a currency union would not be strongly advised.

Cerqueira (2013) to detect a potential change in the cross-country core-periphery pattern. While the former are standard and widely used in economics, the nonparametric measures in form of local polynomial regressions (LPR) have the advantage that they are very flexible in their use and do not rely on prior assumptions about a particular functional form regarding the co-movement of economic activity. Thus, our quarterly correlation index and the LPR framework permit us to specifically track the changing relationship over time. Since our empirical investigation covers the time period from 1970Q1 to 2015Q4, we are able to investigate the potential (de)synchronizing effects of both the great financial crisis and the subsequent sovereign debt crisis, an issue that only a limited number of studies have addressed so far (Grigoraş and Stanciu, 2016; Degiannakis *et al.*, 2014). Apart from the synchronization of business cycles, we also study their amplitudes because different amplitudes can lead to diverging cyclical conditions even if the cycles are perfectly correlated.

These aspects relate to the future of the economic governance of the EMU, since the negative effects of strongly diverging business cycles could be minimized by common institutions for risk sharing. De Grauwe and Ji (2016), for instance, argue that risk-sharing efforts to stabilize the business cycles should be strengthened relative to the efforts that have been taken to conduct structural reforms. This is because they show empirically that a large part of the divergence in the Eurozone was the result of business cycle movements whose amplitudes differ across countries.

Our main empirical findings can be summarized as follows: both our panel and LPR estimations suggest that the output co-movement between core and peripheral countries decreased markedly in the wake of the financial crisis. Analyzing the synchronization between core and peripheral economies among themselves, our results point to a rather stark drift among peripheral countries around the financial crisis, which was partly reversed during the sovereign debt crisis. Core economies, on the other hand, enjoyed a rising synchronization of output during both the financial and the subsequent sovereign debt crisis. We observe the same pattern when comparing the alignment of economic activity of our two clusters with countries outside of the EMU: the drop in synchronization is significantly more pronounced for countries belonging to the periphery than to the core. We also find that there are large differences in the amplitudes of national cycles, which seem to be related to the extent to which national economies react to the common cycle. But these differences are not systematically related to the core-periphery divergence in correlations. This suggests that the core-periphery desynchronization is not the only problem for the euro area.

The remainder of this article proceeds as follows: Section 2 gives a brief overview of the related literature, Section 3 discusses our data, Section 4 describes our empirical methods and the estimation results, while Section 5 concludes.

## 2. Related Literature

The debate about whether and how business cycles are synchronizing in the EMU has been the focus of an intense academic exchange. In this section, we will give a short and necessarily selective review of the vast literature relevant in our context.<sup>4</sup>

Before the start of the EMU, several studies examined whether the economies that planned to form a common currency area in Europe could be considered an optimum currency area. Bayoumi and Eichengreen (1992) use structural vector autoregressions (SVAR) to identify the incidence of supply and demand shocks in Europe. Both types of shocks are estimated to be smaller in magnitude and more intercorrelated for a group of core countries, consisting of Germany, France, Belgium, the Netherlands and Denmark - than for the other European Community (EC) countries, such as Italy, Spain, Portugal, Ireland, Greece and the UK. Overall, "there is also little evidence of convergence in the sense of the core-periphery distinction becoming less pronounced over time" (Bayoumi and Eichengreen, 1992, p. 34). In the same vein, Dickerson *et al.* (1998) find a clear difference in business cycles between a group of core countries (Netherlands, France, Belgium, Luxem-

<sup>&</sup>lt;sup>4</sup>While we focus on studies that investigate business cycle synchronization at the national level, another strand of the literature focus on co-movement between regions in the EU. See, for instance, Anagnostou *et al.* (2015), Bierbaumer-Polly *et al.* (2016) or Belke and Heine (2006).

bourg, Germany) and the periphery (Greece, Ireland, Italy, Portugal, Denmark, United Kingdom, Spain) prior to the implementation of the EMU.

Christodoulakis *et al.* (1995) tend to have a more optimistic view. According to their empirical framework, countries of the EC react remarkably similar to shocks, even though the nature of the shocks might be rather different. Mostly, these differences were found to be related to institutions (e.g. labour market regulations) or policy variables (such as government consumption). As a result, "observed differences in shocks and business cycles mechanisms will tend to melt down as common institutions and politics start to emerge" (Christodoulakis *et al.*, 1995, p. 16), a view shared by the European Commission (1990) in its seminal study *One Market, One Money.* Moreover, they do not find any evidence for a core-periphery distinction. Others point to the increase of business cycle synchronization in the run-up to the EMU (Angeloni and Dedola, 1999; Fatas, 1997).

Would the introduction of the Euro spur business cycle synchronization among member states? Proponents of the endogeneous OCA analysis, based on the seminal papers of Frankel and Rose (1997, 1998), tended to agree. They argued that differences in the co-movement of aggregate output between future members of the EMU were not as problematic as initially claimed, since member states would meet the necessary OCA criteria better after the introduction of the common currency, i.e. ex post than ex ante. Following this logic, the introduction of the EMU would have led to a marked increase in economic and financial integration that would consequently help to align economic fluctuations among member states. With the benefit of hindsight, the empirical evidence is conflicting, making the notion of whether the introduction of EMU led to higher business cycle synchronization contested.<sup>5</sup>

Studies detecting a positive effect include Furceri and Karras (2008). They use a simple pairwise correlation coefficients of GDP and unemployment and find that all 12 countries in their sample are better synchronized with the common EUwide cycle after the start of the EMU. The positive effects of EMU on business

<sup>&</sup>lt;sup>5</sup>For a recent survey about business cycle synchronization in Europe, see de Haan *et al.* (2008). Willett *et al.* (2010) provide an overview of the endogeneous OCA theory and its validity for the first years of the EMU.

cycle synchronization are confirmed, with albeit different statistical methods, by Altavilla (2004), Gogas (2013), Darvas and Szapáry (2008), Gonçalves *et al.* (2009), among others, as well as for different proxies for economic activity such as industrial production (Gayer, 2007) or economic sentiment indicators (Aguiar-Conraria *et al.*, 2013). On the other hand, other studies fail to detect any EMU effect on business cycle synchronization (Camacho *et al.*, 2006; Weyerstrass *et al.*, 2011), or find even a dampening one (Papageorgiou *et al.*, 2010).<sup>6</sup>

Another string of literature suggests rather heterogeneous effects of joining the EMU across member states. Lehwald (2013) uses a dynamic factor model and compares the common European factor for the initial EA-12 before and after the introduction of the EMU. The results suggest that the implementation of the EMU had a dampening effect for the peripheral economies, defined as Spain, Portugal, Greece and Ireland, but spurred the co-movement for the core (i.e. the other 8 economies). Along the same lines, Papageorgiou *et al.* (2010) argue that Europe tended to converge during the period 1992 - 1999, but to diverge from 2000 - 2009 on, leading to an increasing number of clusters within Europe. Konstantakopoulou and Tsionas (2011) use a dynamic analysis based on the autoregressive distributed lag model for the time-span of 1960-2009. Their results point to a core group of countries, such as Germany, the Netherlands, Austria, France and Belgium that are highly synchronized among themselves, while especially Greece and Portugal do not show any synchronization with the other economies, corroborating the results of Gouveia and Correia (2008).<sup>7</sup>

The tumultuous period of the financial crisis and the following sovereign debt crisis brought deep economic imbalances within the EMU to the limelight (Lane, 2012). However, there are only a very limited number of studies focusing on the potential (de)synchronization of economic activity during this period. Degiannakis *et al.* (2014) investigate the time-varying correlation between the EU-12 and its initial members using scalar-BEKK, a widely used model of conditional covariances and correlations, and a multivariate riskmetrics framework. While business

<sup>&</sup>lt;sup>6</sup>Generally, it is not easy to separate the effects of the introduction of the Euro from other EU-wide initiatives, as pointed out by Willett *et al.* (2010).

<sup>&</sup>lt;sup>7</sup>However, the core-periphery dualism seems to be less pronounced in Europe than in the United States, according to the estimates of Ferreira-Lopes and Pina (2011).

cycles in the EMU were getting increasingly correlated until 2007, the financial crisis triggered a desynchronization process, most prominently for Greece, Ireland, Portugal and Spain, but also for non-EMU countries such as the United Kingdom or Sweden, supporting the analysis of Gächter *et al.* (2012) who find a decrease in correlation and an increase in the dispersion of synchronization levels due to the financial crisis within the EMU. Grigoraş and Stanciu (2016) compare concordance and correlation measures for a sample of European economies, concluding that the co-movement of economic activity among European economies diverged after the outbreak of the crisis (starting 2009Q1) compared to the Euro adoption period (2002Q1 - 2008Q4). Ferroni and Klaus (2015) use a factor model to determine the business cycle properties of the four largest European economies during the recent European debt crisis. They find that Germany, France and (surprisingly) Italy are well-aligned with the EA cycle, whereas Spain shows an asymmetric behaviour.

Differences in the results can be partly explained by the myriad of empirical methods to determine the level of business cycle synchronization: for instance coherence and concordance measures based on business cycle dating algorithms (Harding and Pagan, 2002, 2006; Artis *et al.*, 2004; Grigoraş and Stanciu, 2016), dynamic factor models (Lee, 2013; Lehwald, 2013; Kose *et al.*, 2003), dynamic correlations (Croux *et al.*, 2001; Fidrmuc and Korhonen, 2010), rolling coefficients (Gayer, 2007), correlation coefficients (Furceri and Karras, 2008) or wavelet analysis (Aguiar-Conraria and Soares, 2011), among others. For our empirical analysis, we employ three different methods to analyze whether business cycles have diverged between the core and the periphery: (i) correlation coefficients as most prominent measure to assess business cycle synchronization in the literature (Bordo and Helbling, 2003), (ii) panel regressions that allow for a more systematic representation and (iii) nonparametric regressions that are specifically useful to track non-linear dynamics over time.

## 3. Data and Variables

### 3.1. Data

For our empirical analysis, we use the seasonally adjusted real gross domestic product (GDP) on a quarterly basis from the OECD. Our dataset includes all member states of the euro area-12 (EA-12)<sup>8</sup> plus Norway, Switzerland, Denmark and Sweden as non-EMU members, with data ranging from 1970Q1 to 2015Q4.

We focus on gross domestic product as our main indicator for business cycle movements for two reasons: it is (i) the most comprehensive measure for aggregate economic activity and (ii) the most widely used and accepted measure in the academic literature and in the general public (Haan *et al.*, 2008; Grigoraş and Stanciu, 2016).<sup>9</sup>

### 3.2. Calculating Business Cycles

Business cycle refers to the movement of economic activity over time (Burns and Mitchell, 1946). Generally, two different types of cycles can be distinguished: the classical and the growth cycle. In the spirit of Lucas (1977), we focus on the growth cycle, i.e. the evolution of the cyclical component of GDP around its long-term trend instead of the classical cycle that is focused on the fluctuations of aggregate economic activity.<sup>10</sup> The literature proposes several different methods to separate the long-term growth trend from the cyclical component (Weyerstrass *et al.*, 2011). For our study, we employ the Hodrick-Prescott (HP) filter (Hodrick

<sup>&</sup>lt;sup>8</sup>EA-12 includes the eleven countries that adopted the Euro in 1999 (Austria, Belgium, Germany, Spain, Finland, France, Ireland, Italy, Luxembourg, the Netherlands and Portugal) as well as Greece that joined in 2001.

<sup>&</sup>lt;sup>9</sup>Alternatively, other potential measures include industrial production (Imbs, 2010) or the unemployment rate (Fatas, 1997). On the upside, these indicators are available at higher frequencies. On the downside, they are not as comprehensive a measure for aggregate activity as GDP, and - in the case of the unemployment rate - depend heavily on country-specific institutions that make cross-country comparisons difficult.

<sup>&</sup>lt;sup>10</sup>From a theoretical point of view, the output gap seems to be the relevant indicator for central banks to monitor (Gächter and Riedl, 2014). From a technical point, our correlation index, as explained in further detail in Section 3.3, requires covariance-stationary time-series.



Figure 1: Business Cycle Fluctuations over time

Notes: Cyclical component of business cycles, using the Hodrick-Prescott filter with  $\lambda$ =1600.

and Prescott, 1997) to detrend the national real GDP series. Belke and Heine (2006) show that the HP filter produces cyclical components broadly similar to those obtained by the bandpass filter. The HP filter is not without criticism,<sup>11</sup> but due to its simple estimation and implementation it remains widely used in the business cycle literature. The cyclical component corresponds to the output gap, whereas the trend can be interpreted as potential output. We follow the literature and use a smoothing parameter of  $\lambda$ =1600 for quarterly data (Ravn and Uhlig, 2002). We depict the cyclical movement of the core and peripheral countries in Figure 1.

 $<sup>^{11}{\</sup>rm See},$  for instance, Canova (1998).

#### 3.3. Correlation Index

In the following, we introduce a correlation index developed by Cerqueira and Martins (2009) that gains increasing popularity in business cycle research (Gächter and Riedl, 2014; Bierbaumer-Polly *et al.*, 2016). As opposed to normal correlation coefficients, the correlation index of Cerqueira and Martins (2009) provides a measure of bilateral correlation for higher - in our case quarterly - frequencies and captures time-variability in bilateral business cycle correlations since it distinguishes between specific episodes of higher and lower synchronization, respectively. This correlation index can be applied to any two covariance-stationary time-series  $g_{it}$ and  $g_{jt}$ :<sup>12</sup>

$$\rho_{ijt} = 1 - \frac{1}{2} \left( \frac{g_{it} - \overline{g_i}}{\sqrt{\frac{1}{T} \sum_{t=1}^{T} (g_{it} - \overline{g_i})^2}} - \frac{g_{jt} - \overline{g_j}}{\sqrt{\frac{1}{T} \sum_{t=1}^{T} (g_{jt} - \overline{g_j})^2}} \right)^2 \tag{1}$$

When averaged over time, the correlation index is identical to the conventional linear correlation coefficient  $\rho_{ij}$ , i.e.  $\rho_{ij} = \frac{1}{T} \sum_{t=1}^{T} \rho_{ijt}$ . However, the index in equation 1 suffers from an asymmetric range: it is bounded between 3-2T and 1 and thus more dispersed over negative than over positive values. Cerqueira (2013) proposes the following transformation

$$\rho_{ijt}^{nb} = \frac{1}{2} ln \left( \frac{1 + \frac{\rho_{ijt}}{2T - 3}}{1 - \rho_{ijt}} \right)$$
(2)

which we will use for our subsequent estimations. Another way to examine the co-movement between country pairs are so-called "rolling window" correlation coefficients, an approach widely used in the business cycle literature (Weyerstrass *et al.*, 2011). In contrast to rolling window correlation coefficients, the proposed correlation index provides distinct advantages: first, there is no loss of observations, second, there is no need to set a an arbitrary window-length, and, third, it is easier

<sup>&</sup>lt;sup>12</sup>Augmented Dickey Fuller tests reject the null hypothesis of a unit root process at conventional significance levels for all series.

to implement in econometric analysis since rolling-window correlations suffer from heavy autocorrelation and hence heavily autocorrelated dependent variables.<sup>13</sup>

# 4. Empirical Framework and Results

## 4.1. Pairwise Correlation Coefficients

As a benchmark analysis, we start our empirical investigation with the estimation of simple correlation coefficients which correspond to the average of our (unbounded) correlation index  $\rho_{ijt}$  over the same sample period. According to this measure, national cycles between two countries are synchronized if both are positively and statistically correlated with each other. In order to analyze time variation in the synchronization of business cycles, we have divided our data in two non-overlapping time periods and calculate the correlation coefficient for each time period separately: the first period starts in 1999Q1 and lasts until 2007Q4, corresponding to the introduction of the Euro until the beginning of the financial crisis that started in the US, following Bekiros *et al.* (2015), while the second period covers the phase after the financial crisis erupted (2008Q1 - 2015Q4). We present our results in three tables, focusing on (i) individual core countries (Table 1), (ii) individual periphery countries (Table 2) and (iii) countries outside the EMU (Table 3).

As depicted in Table 1, core countries tend to have quite a high correlation between each other as well as with the EA-12<sup>14</sup> and the periphery aggregate in the pre-crisis period. The Netherlands and Finland show the lowest correlation with the periphery and the EA-12, and Austria and Germany the highest, respectively. The correlation between individual core countries and the periphery starts to drop after the start of the financial crisis. The drop is most pronounced for Austria and

<sup>&</sup>lt;sup>13</sup>For a further discussion, see also Gächter and Riedl (2014) and Degiannakis *et al.* (2014).

<sup>&</sup>lt;sup>14</sup>Naturally, correlation coefficients between EMU member states and the EA-12 are biased upwards. This bias is more pronounced for countries that have a higher weight in the EA-12 aggregate (Haan *et al.*, 2008).

France, where the correlation coefficient decreased from 0.934 (Austria) and 0.927 (France) to 0.619 and 0.642.

	Finland	France	Germany	Austria	Netherlands	Periphery	EA-12
Pre-Crisis							
Finland	1						
France	$0.894^{***}$	1					
Germany	$0.780^{***}$	$0.859^{***}$	1				
Austria	$0.852^{***}$	$0.928^{***}$	$0.854^{***}$	1			
Netherlands	$0.838^{***}$	$0.903^{***}$	$0.856^{***}$	$0.902^{***}$	1		
Periphery	$0.866^{***}$	$0.927^{***}$	$0.928^{***}$	$0.934^{***}$	$0.864^{***}$	1	
EA-12	$0.879^{***}$	$0.953^{***}$	$0.963^{***}$	$0.942^{***}$	$0.914^{***}$	$0.984^{***}$	1
Crisis							
Finland	1						
France	$0.942^{***}$	1					
Germany	$0.971^{***}$	$0.966^{***}$	1				
Austria	$0.927^{***}$	$0.951^{***}$	$0.948^{***}$	1			
Netherlands	$0.872^{***}$	$0.771^{***}$	$0.827^{***}$	$0.793^{***}$	1		
Periphery	$0.719^{***}$	$0.642^{***}$	$0.679^{***}$	$0.619^{***}$	$0.906^{***}$	1	
EA-12	$0.946^{***}$	$0.917^{***}$	$0.943^{***}$	$0.895^{***}$	$0.938^{***}$	$0.881^{***}$	1

Table 1: Correlation coefficients: core (individual) vs. periphery (aggregate)

<u>Notes</u>: The cyclical component is extracted using the HP filter for the logarithmic real and seasonally adjusted quarterly GDP. The "pre-crisis" period is defined as ranging from 1999Q1 - 2007Q4, the "crisis" period from 2008Q1 - 2015Q4. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

For the periphery (see Table 2), the general correlation between these countries with the EA-12 and the core seems less pronounced. Italy and Spain show the highest, Portugal and Greece the lowest correlation with economic activity in the core and the aggregate EA-12. The peripheral economies decreased their degree of synchronization with the euro area and the core in the crisis period, with the exception of Italy, with Spain and Greece showing the most pronounced drop. In general, Greece seems to have a rather low degree of synchronization with other European economies, an observation already pointed out by Gayer (2007): it shows a moderate (positive) but insignificant correlation with the EA-12 but negative (however insignificant) correlation with core countries in the crisis period. Interestingly, Italy enjoyed in both sample periods a very strong co-movement of economic activity with both the core and the EA-12 aggregate.

	Italy	Portugal	Greece	Ireland	Spain	Core	EA-12
$\underline{Pre-Crisis}$							
Italy	1						
Portugal	$0.603^{***}$	1					
Greece	$0.544^{***}$	0.261	1				
Ireland	$0.814^{***}$	$0.613^{***}$	$0.488^{***}$	1			
Spain	$0.884^{***}$	$0.751^{***}$	$0.600^{***}$	$0.861^{***}$	1		
Core	$0.911^{***}$	$0.801^{***}$	$0.552^{***}$	$0.830^{***}$	$0.958^{***}$	1	
EA-12	$0.941^{***}$	$0.772^{***}$	$0.587^{***}$	$0.858^{***}$	$0.969^{***}$	$0.994^{***}$	1
Crisis							
Italy	1						
Portugal	$0.677^{***}$	1					
Greece	0.0615	$0.612^{***}$	1				
Ireland	$0.727^{***}$	$0.549^{***}$	$0.341^{*}$	1			
Spain	$0.676^{***}$	$0.822^{***}$	$0.643^{***}$	$0.816^{***}$	1		
Core	$0.917^{***}$	$0.416^{**}$	-0.114	$0.658^{***}$	$0.502^{***}$	1	
EA-12	$0.967^{***}$	$0.628^{***}$	0.130	$0.785^{***}$	$0.719^{***}$	$0.958^{***}$	1

Table 2: Correlation coefficients: periphery (individual) vs. core (aggregate)

Notes: The cyclical component is extracted using the HP filter for the logarithmic real and seasonally adjusted quarterly GDP. The "pre-crisis" period is defined as ranging from 1999Q1 - 2007Q4, the "crisis" period from 2008Q1 - 2015Q4. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

One way to test whether the membership of the euro area influenced the way the financial crisis had an impact on business cycle co-movements is to examine the correlations also for countries that are outside of the EMU but with economies very much integrated with the euro area. We therefore analyze the correlation patterns for four non-EMU countries: Denmark, Sweden (EU member countries who have opted to stay out of the EMU) as well as Switzerland and Norway (as part of EFTA). Denmark has opted for a fixed exchange rate to the euro and should thus, from a macroeconomic point of view, be regarded as an "informal" member of the core. The other three countries have a flexible exchange rate regime (to a varying degree) with Norway the added specificity of its reliance on oil revenues, which should make its economy less dependent on the euro area cycle. Results are depicted in Table 3. In the pre-crisis period, there is no stark difference between the correlations of the core and the periphery with the non-EMU countries. In both cases, Switzerland had the highest correlation and Norway the lowest. Yet, the co-movement patterns changed after the eruption of the financial crisis. Compared

	Denmark	Sweden	Norway	Switzerland	Periphery	Core	EA-12
Pre-Crisis							
Denmark	1						
Sweden	$0.903^{***}$	1					
Norway	$0.692^{***}$	$0.732^{***}$	1				
Switzerland	$0.816^{***}$	$0.812^{***}$	$0.653^{***}$	1			
Periphery	$0.831^{***}$	$0.893^{***}$	$0.680^{***}$	$0.929^{***}$	1		
Core	$0.793^{***}$	$0.849^{***}$	$0.639^{***}$	$0.948^{***}$	$0.961^{***}$	1	
EA-12	$0.818^{***}$	$0.878^{***}$	$0.667^{***}$	$0.951^{***}$	$0.984^{***}$	$0.994^{***}$	1
Crisis							
Denmark	1						
Sweden	$0.879^{***}$	1					
Norway	$0.388^{**}$	0.201	1				
Switzerland	$0.867^{***}$	$0.734^{***}$	$0.448^{**}$	1			
Periphery	$0.736^{***}$	$0.773^{***}$	0.278	$0.671^{***}$	1		
Core	$0.926^{***}$	$0.899^{***}$	$0.379^{**}$	$0.876^{***}$	$0.709^{***}$	1	
EA-12	$0.921^{***}$	$0.917^{***}$	$0.368^{**}$	$0.862^{***}$	$0.881^{***}$	$0.958^{***}$	1

Table 3: Correlation coefficients: outside EMU countries vs. aggregates

Notes: The cyclical component is extracted using the HP filter for the logarithmic real and seasonally adjusted quarterly GDP. The "pre-crisis" period is defined as ranging from 1999Q1 - 2007Q4, the "crisis" period from 2008Q1 - 2015Q4. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

to the pre-crisis period, all non-EMU countries have less synchronization vis-à-vis the periphery, while correlation with the core stayed more or less at the same level. Norway seems like an exception since output synchronization fell significantly for both the core and the periphery. But this might be due to the high variability of oil prices after 2007/8. All in all it appears that the reaction of these European economies to the crisis was not visibly affected by their non-membership to the euro area. The sovereign debt crisis does not seem to have affected the relatively close correlation between the core countries and these other economies not part of the euro area. It seems that the relatively strong external position of the non-euro countries, similar to the one of the core euro countries, was more important than the fact that they are not part of the EMU.

At the aggregate level, our analysis suggests that correlation between the core and the periphery fell, as one would expect, after the eruption of the financial crisis. The non-EMU countries, with their strong external positions, did not experience risk spreads and showed a lower synchronization with the periphery in the crisis period, but no significant change in the output co-movement with the core countries.

#### 4.2. Panel Regression

In this section, we run variants of the following panel regression to test whether the eruption of the global financial crisis changed the business cycle synchronization on a country-pair level:

$$\rho_{ijt}^{nb} = \beta_0 + \beta_1 Cluster_{ij} \times Crisis_t + \delta_i + \gamma_j + \alpha_{ij} + \omega_t + \epsilon_{ijt} \tag{3}$$

We regress our quarterly correlation index  $\rho_{ijt}^{nb}$  (see equation 2) on an interaction term between a crisis dummy that equals 1 starting from 2008Q1 and 0 otherwise and four different country clusters: core, periphery, non-EMU and EA-12. A statistically significant value for the coefficient  $\beta_1$  in equation 3 indicates a structural shift in the business cycle synchronization among the specific clusters. We include all our EA-12 plus four non-EMU countries (Norway, Sweden, Denmark, Switzerland) and run our regressions from the start of the EMU in 1999Q1 until 2015Q4. Furthermore, we include time, country and bilateral fixed effects to control for unobserved heterogeneity.

Results are depicted in Table 4 for the core and periphery (i) among and between themselves (column 1), (ii) vis-à-vis the EA-12 (column 2) and (iii) relative to the non-EMU countries (column 3), defined as Denmark, Sweden, Norway and Switzerland. Note that we do not include dummies for the crisis and the clusters since they are absorbed by the time and country-pair fixed effects.

Our estimations support our initial findings based on correlation coefficients: the negative interaction coefficient suggests that countries along the core-periphery dimension were less aligned after the eruption of the financial crisis (column 1). This effect is significant at the 1% level. Interestingly, while core countries were growing stronger together as a group, peripheral countries were drifting apart. As a result, the core countries grew stronger together as a block among themselves, the pe-

	(1)	(2)	(3)
Crisis $\times$ Core - Core	$0.205^{**}$ (0.0880)		
Crisis $\times$ Per Per.	$-0.252^{***}$ (0.0880)		
Crisis $\times$ Core - Per.	$-0.535^{***}$ (0.0604)		
Crisis $\times$ Core - EA-12		-0.00694 (0.0516)	
Crisis $\times$ Per EA-12		$-0.405^{***}$ (0.0516)	
Crisis $\times$ Core - non-EMU			$\begin{array}{c} 0.181^{***} \\ (0.0657) \end{array}$
Crisis $\times$ Per non-EMU			$-0.204^{***}$ (0.0657)
Observations	8159	8159	8159
$R^2$	0.159	0.156	0.151
Time FE	$\checkmark$	$\checkmark$	$\checkmark$
Country FE	$\checkmark$	$\checkmark$	$\checkmark$
Bilateral FE	$\checkmark$	$\checkmark$	$\checkmark$
Sample Mean $\rho^{nb}$	1.320	1.320	1.320

Table 4: Business cycle synchronization

Notes: Dependent variable is the unbounded quarterly bilateral correlation index by Cerqueira (2013). Clustered standard errors at the country-pair level in parentheses. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level. Specifications (1) - (3) include country, bilateral and time (quarterly) fixed effects. The dummy variable "crisis" equals 1 starting from 2008Q1.

ripheral countries as cluster were less aligned relative to the core, but also relative to themselves. Comparing the core and peripheral countries with country-pairs belonging to the EA-12, a similar picture emerges (column 2). Peripheral countries were significantly less aligned with EA-12 countries during the crisis period, whereas core countries did not have a statistically significant change in the synchronization of economic output. A drop in synchronization is also detected for non-EMU countries with regards to the periphery, which stands in contrast with the increased alignment of the non-EMU countries with the core (column 3). Judging from these results, the period after 2007 initiated a decoupling in the Eurozone, and resulted in a group of core countries that over time increased their synchronization of economic activity, both among themselves and towards neighboring countries that are not part of the euro area. On the other side, the peripheral countries decoupled from the core, non-EMU countries and among themselves.

Concerning the "decoupling" within the periphery, all these countries had periods of high risk premia in common. But the root cause of their difficulties was very different: Ireland and Spain had a real estate boom which burst with the financial crisis. But the economies of these two countries started to recover quickly once the banking problems in the real estate sector had been resolved. Portugal and Greece, by contrast, had more fundamental problems of growth and productivity and have not fully recovered even today.

#### 4.3. Local Polynomial Regressions

Both our correlation coefficients and our panel regression point to a split between the synchronization of peripheral and core business cycles after the eruption of the financial crisis. However, our results may hide some important variability given our time aggregation in two non-overlapping periods. Generally, linear models provide a simple fit to the data, are easy to understand and widely used. However, they are less suited if the data shows nonlinearities. In our case, business cycle synchronization across different country groups is likely to fluctuate over time. Nonparametric methods provide a useful modeling alternative since they are widely used in data analysis to determine unknown trends and do not rely on any specific functional form.<sup>15</sup> We estimate the following model to analyze cross-country business cycle synchronization over time:

$$\rho_{ijt}^{nb} = f(t) + u_{ijt} \tag{4}$$

<sup>&</sup>lt;sup>15</sup>For an in-depth treatment, see Fan and Gijbels (1996), especially Chapter 3.

Since f(t) is unknown, we use local polynomial regressions as a smoothing approach that makes no assumptions about the functional form of f(t). This approach models the regression locally in the neighborhood around each point of interest  $t_0$  with a window width of h, using a p-order Taylor expansion as approximation of the true function. In the estimation process, a weighting function K (Kernel) determines how observations in the neighborhood of  $t_0$  are weighted. The quality of the fit of the LPR is affected by these three critical parameters. For our estimations, we use an Epanechnikov kernel, a polynomial of order 1 and rely on a rule-of-thumb (ROT) algorithm to automatically determine the optimal bandwidth, following closely the setting of Cerqueira (2013). We present our LPR results with pointwise confidence intervals (95%) (Cameron and Trivedi, 2005). Contrary to our analysis based on correlation coefficients and panel data, we take advantage of the full length of our sample starting in 1970 in order to track changes of synchronization over an extended time horizon, even though our primary interest remains on potential changes in the recent crisis period.

Results for the core and periphery cluster are depicted in Figure 2. Synchronization between core and peripheral countries was rather low during the 1970s until the beginning of the 1990s (see Figure 2a). During the run-up to the EMU, cycles seemed to synchronize rather sharply, in line with other studies that mention the "Maastricht effect" (Papageorgiou *et al.*, 2010; Inklaar *et al.*, 2008; Aguiar-Conraria and Soares, 2011). After the establishment of the currency area, the output co-movements continued to oscillate on a rather high level in the early 2000s. The process of higher synchronization between the two clusters has come to a hold shortly before the beginning of the financial crisis and consequently before the eruption of the sovereign debt crisis, thus corroborating the findings of Ferroni and Klaus (2015) in the case of Spain. Our findings are also in line with Degiannakis *et al.* (2014) who suggest a general reduction in the co-movement of economic output among peripheral economies and the EMU-12 cycle after 2007.<sup>16</sup>

<sup>&</sup>lt;sup>16</sup>As local polynomial regressions take data before and after  $t_0$  into account, our results anticipate to a certain extent future changes in synchronization. Therefore, as opposed to assigning a certain level of synchronization to a certain date, the LPR results are best interpreted as overall tendencies over a period of time.

Synchronization starts to slightly increase again after 2010, but remains at a low level.<sup>17</sup>



Figure 2: Local polynomial regressions: core and periphery

Notes: Local polynomial regressions for core and peripheral countries.

Starting from the mid-1970s, peripheral countries steadily increased synchronization among themselves that peaked around the mid-1990s (see Figure 2b).

<sup>&</sup>lt;sup>17</sup> In order to check whether our slope is different from a constant, we implement a significance test that is analogous to a simple t-test in a parametric regression setting (Racine, 1997; Racine *et al.*, 2006). For all the nonparametric models presented in this section, the test rejects the notion that our explanatory variable exerts a constant effect at a 1% significance level, i.e. the synchronization between the clusters varies significantly over time.

From there on, the alignment of economic activity hovered around a steady level, which on average was however lower than for the core countries. The co-movement of output decreased among peripheral economics during the financial crisis. However, synchronization started to increase from 2010 on, potentially reflecting the subsequent symmetric impact of the sovereign debt crisis on economic activity.

The relationship among core countries shows a slow, but steady upwards trend since the 1990s (see Figure 2c). The first plateau is reached after the rather strong increase in synchronization after the German unification process, and then in the early years of the EMU. Rather interestingly, and opposed to the behaviour of peripheral countries, there seems to exist no real dampening impact of either the financial crisis or the sovereign debt crisis on the alignment of economic activity among core countries. Rather to the contrary, economic synchronization has had a slight upwards tick in the period after 2010.

Figure 3 shows the development between the core and periphery vis-à-vis countries outside of the EMU. These economies enjoyed a rather low level of synchronization with all the EA-12 countries up until 1990, but then gradually increased to synchronize up until the mid-2000 (Figure 3a). The onset of the financial crisis triggered a rapid decrease in correlation, corroborating the results of Degiannakis *et al.* (2014) that non-EMU countries decoupled from the EMU-cycle in 2007. However, the synchronization of economic activity between non-EMU and EA-12 countries started to increase again around 2010. Splitting the EA-12 sample in the periphery (Figure 3b) and core clusters (Figure 3c), it becomes obvious that the most recent slowdown in synchronization is particularly driven by the periphery.



Figure 3: Local polynomial regressions: non-EMU countries

Notes: Local polynomial regressions for non-EMU countries relative to core, peripheral and EA-12 countries.

Overall, our dynamic analysis relying on local polynomial regressions confirms our previous results. Core and peripheral countries uncoupled significantly in the wake of the financial crisis. It also confirms that core countries seem to enjoy a higher (albeit on a very low level) synchronization after the crisis. Turning to the correlation among countries of the periphery, the financial crisis led first to asymmetric behavior, but turned into more synchronization due to a common shock in form of the sovereign debt crisis. Economic activity with non-EMU countries was particularly uncoupled for peripheral countries around the financial crisis, but less so for the core.



Figure 4: Local polynomial regressions: robustness check

Notes: Local polynomial regression between core and peripheral countries without Italy and Greece, respectively.

As a robustness check, we provide our local polynomial regression estimations with a different set of countries for the periphery cluster. Our analysis in Section 4.1 suggests that both Greece and Italy are potential outliers, with the former having the lowest and the latter the strongest correlation with core countries. As pointed out by Campos and Macchiarelli (2016), the periphery pattern that emerged after the introduction of the Euro included Spain, Portugal, Ireland and Greece, but excluded Italy. Other studies, as Lehwald (2013), do not include Italy in their definition of peripheral countries either. Greece, on the other hand, seemed to have rather different output co-movements compared with the rest of the euro area (Gayer, 2007).

Thus, we repeat our local polynomial regressions with the same set of core countries, but exclude consecutively either Italy or Greece in the group of peripheral countries. Results from this exercise are depicted in Figure 4. Compared to the solid line, which represents our initial core-periphery country grouping, the group without Greece is actually more synchronized with the core, pointing to the particular decoupling of the Greek economy with the core countries, which was already taking place shortly after the introduction of the Euro. Our estimations support the findings of Gouveia and Correia (2008) who point out that the output co-movements of Greece were becoming weaker after 1997. If we exclude Italy from the periphery cluster, the desynchronization of the periphery with regards to the core is even more pronounced, suggesting that Italy enjoys on average a higher synchronization with the core than other peripheral countries. Our findings thus support the analysis of Campos and Macchiarelli (2016) who find that Italy may not be directly regarded as peripheral country from a business cycle synchronization point of view.

#### 4.4. Amplitude vs Co-Movement

Most of the literature has focused on co-movements in the cycle as a measure of synchronization. But this can be misleading if the amplitudes of the cycle are very different. Figure 5 illustrates this with an example: there are two countries, which share the same (highly stylized) business cycle, but the amplitudes are very different. This has two implications: the correlation coefficient between the two series is 1, but at the peak and through of the cycle large differences appear. These differences can lead to similar problems of common policy making as if the two cycles were uncorrelated. But the problems posed by different amplitudes appear mostly around the extremes of the cycle. The widespread feeling during the peak of the crisis that the ECB's policy did not fit the needs of any country<sup>18</sup> could thus have two origins: either the correlation pattern changed since the financial crisis or the amplitudes of the cycles were different across countries so that one would expect at any peak and bottom of the cycle similar problems in business cycle positions. Moreover, differences in the amplitude could magnify the impact of any desynchronization.

<sup>&</sup>lt;sup>18</sup>For a theoretical analysis, see Gros and Hefeker (2002).





Notes: This figure shows stylized business cycles for country A and B. The business cycles are synchronous, with country B having an amplitude which is twice as large as country A's.

These considerations suggest that the emphasis on correlations makes sense only if countries have cycles which might be shifted, but have a similar amplitude (and length). Figure 6 depicts the case of two countries whose cycles are shifted over time in such a way that the correlation is equal to zero. In this case the differences in the cyclical positions are minimal at half cycle and remain constant between the peak and trough.

Different amplitudes exacerbate any lack of correlation as shown in Figure 7 which shows three lines: the common euro area cycle, the cycle of a country which is delayed by half a cycle, but has the same amplitude as the common cycle, and a country with an amplitude twice as large as the common cycle. It is apparent that the difference between the national and the euro area cycle is not the same when the amplitude changes although the correlation coefficient

Figure 6: Zero correlation because of phase shift



Notes: This figure shows stylized business cycles for country A and C with the same amplitude but that are uncorrelated.

between the national and the euro area cycle is equal (to zero) in both cases. At the bottom of the last crisis it had been observed that the ECB policy was "one-size-fits-none" as the ECB's policy appeared too loose for the core and too tight for the periphery. A comparison between these figures suggests one way one could discriminate between the two hypotheses: if differences in amplitudes are the problem one would expect large divergences to appear mainly at the peak and trough of the cycle. By contrast, if business cycles are shifted, the divergences in cyclical positions should persist most of the time. The way the policy discussion has evolved suggests that different amplitudes might have been a key factor since differences in cyclical positions play much less of a role today than at the peak of the crisis.





Notes: This figure shows the business cycle of the euro area, country A which is shifted and country B which is shifted and has an amplitude twice as large as country A's.

A first measure of different amplitudes is the volatility of business cycles, at the national level, proxied by the standard deviation of the cyclical component of national real GDP series. A first question is thus whether there has been convergence in the amplitude of business cycles measured by their overall volatility (Gayer, 2007). As before, we rely on the HP filter for the trend-cycle decomposition.

As in previous sections, we are interested in the effects of the potential changes since the start of the financial crisis. Thus, we split the sample between the early phase of the Euro (1999Q1-2007Q4) and the crisis period (2008Q1-2015Q4), but also provide the standard deviation for the whole period from 1999Q1 to 2015Q4. The results are depicted in Table 5.

1999Q1-2015Q4	1999Q1-2007Q4	2008Q1-2015Q4
1.32	1.12	1.52
1.39	1.24	1.53
1.29	1.18	1.40
2.47	1.47	3.27
1.39	1.17	1.60
1.40	1.07	1.70
2.56	2.49	2.50
1.43	1.24	1.63
2.11	1.76	2.43
1.03	0.95	1.07
1.70	1.48	1.92
1.34	1.20	1.47
1.46	1.52	1.40
1.12	1.19	1.04
1.83	1.50	2.09
1.52	1.50	1.50
1.27	1.42	1.10
	$\begin{array}{c} 1999 \text{Q1-2015} \text{Q4} \\ \hline 1.32 \\ 1.39 \\ 1.29 \\ \hline 2.47 \\ 1.39 \\ 1.40 \\ 2.56 \\ 1.43 \\ \hline 2.56 \\ 1.43 \\ \hline 1.12 \\ 1.34 \\ 1.46 \\ \hline 1.12 \\ 1.83 \\ 1.52 \\ 1.27 \\ \hline \end{array}$	$\begin{array}{c ccccc} 1999Q1-2015Q4 & 1999Q1-2007Q4 \\ \hline 1.32 & 1.12 \\ 1.39 & 1.24 \\ 1.29 & 1.18 \\ \hline \\ 2.47 & 1.47 \\ 1.39 & 1.17 \\ 1.40 & 1.07 \\ 2.56 & 2.49 \\ 1.43 & 1.24 \\ \hline \\ 2.11 & 1.76 \\ 1.03 & 0.95 \\ 1.70 & 1.48 \\ 1.34 & 1.20 \\ 1.46 & 1.52 \\ \hline \\ 1.12 & 1.19 \\ 1.83 & 1.50 \\ 1.52 & 1.50 \\ 1.27 & 1.42 \\ \hline \end{array}$

Table 5: Amplitude of business cycles

Notes: Standard deviations of output gap in %.

Focusing on the development of volatility in the periods after the introduction of the Euro and before the crisis (column 2), Ireland and Greece exhibit the highest business cycle volatility in the periphery, which is also above the average of the (aggregate) EA-12. Among the core countries, Finland, the Netherlands and surprisingly Germany stand out, whereas France shows the lowest variation. In the period after 2007Q4, the volatility of national business cycles was generally on the rise for all peripheral and core countries, except for the Netherlands. Yet, Greece and Ireland had the highest volatility, closely followed by Finland. While the variation across countries is rather pronounced, it does not run along the core-periphery dimension. For the functioning of the euro area, the more relevant question is whether national cycles of individual countries had a different reactivity to the aggregate EA-12 cycle, i.e whether different countries had different "betas". To measure these "betas", we run the following simple regression with heteroskedasticity- and autocorrelation-consistent (HAC) standard errors in the spirit of Gogas (2013) and De Grauwe and Ji (2016):

$$ln(c_t^i) = \beta_0 + \beta_1 \times ln(c_t^{EA-12}) + e_t \tag{5}$$

This is done for our core and peripheral countries individually with a sample running from 1999Q1-2015Q4, with  $c_t^i$  being the cyclical component of GDP for each country i and  $c_t^{EA-12}$  for the EA-12, respectively. Results for the core and the periphery are shown in Table 6 and 7.<sup>19</sup>

Table 6: Contemporaneous effect of EA-12 cycle on national cycle: periphery

	Greece	Italy	Spain	Ireland	Portugal
EA-12	0.477	1.031***	0.873***	1.627***	0.758***
	(0.477)	(0.0528)	(0.150)	(0.192)	(0.141)
$R^2$	0.062	0.912	0.650	0.670	0.472

Notes: Dependent variable is the quarterly (HP filtered) cyclical component of each peripheral country. Sample from 1999Q1 - 2015Q4. HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

Table 7: Contemporaneous effect of EA-12 cycle on national cycle: core

	Finland	France	Germany	Austria	Netherlands
EA-12	$\frac{1.499^{***}}{(0.127)}$	$\begin{array}{c} 0.741^{***} \\ (0.0556) \end{array}$	$\frac{1.254^{***}}{(0.0879)}$	$0.949^{***} \\ (0.0547)$	$\frac{1.038^{***}}{(0.0969)}$
$R^2$	0.843	0.868	0.904	0.840	0.841

<u>Notes</u>: Dependent variable is the quarterly (HP filtered) cyclical component of each core country. Sample from 1999Q1 - 2015Q4. HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

<sup>&</sup>lt;sup>19</sup>As a robustness check, we exclude for all regressions the reference country i from the EA-12 aggregate to minimize simultaneity issues. This bias should be negligible for smaller economies, but more relevant for bigger countries like France or Germany. The results do not vary significantly, and are reported in the Appendix.

The results suggest that all national cycles are tightly linked to the euro area one (with one exception, Greece). There is no distinct core-periphery pattern, except for the special case of Greece. Compared to the other nine countries that react positively to a reaction of the aggregate cycle, the coefficient for Greece is not only lower in absolute magnitude, but also statistically not significant.

Within the core there is a large, statistically significant difference between Germany and France: Germany's beta is at 1.25 almost twice as large as that of France (0.74). This means that if the common cycle shows an output gap of 2 %, that of France (compared to its own cycle) would be 1.48 %, whereas that of Germany would be 2.5 %. Any large common up- or downswing could thus lead to very large cyclical positions. Within the periphery one finds similar pronounced differences, with Ireland having a beta of 1.63, compared to 0.87 for Spain. Italy is the only large country with a beta almost exactly equal to one. From the non-EMU countries, Norway seems to react the least to the aggregate EA-12 cycle, having both the lowest coefficient and  $R^2$ , as depicted in Table 8.

Table 8: Contemporaneous effect of EA-12 cycle on national cycle: outside EMU

	Sweden	Denmark	Norway	Switzerland
EA-12	$1.276^{***}$	1.023***	$0.454^{***}$	0.877***
	(0.133)	(0.0786)	(0.114)	(0.0869)
$R^2$	0.807	0.757	0.274	0.789

<u>Notes</u>: Dependent variable is the quarterly (HP filtered) cyclical component of each core country. Sample from 1999Q1 - 2015Q4. HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

To investigate whether our relationship between the national and the aggregate cycle changed after the eruption of the financial crisis, we include a dummy variable into our empirical setting which takes the value one for the period after 2007Q4 and zero otherwise. The interaction term between this crisis dummy and the EA-12 cyclical component indicates whether there was a trend shift, as shown in equation 6:

$$ln(c_t^i) = \beta_0 + \beta_1 ln(c_t^{EA-12}) + \beta_2 D_{crisis} + \beta_3 ln(c_t^{EA-12}) \times D_{crisis} + e_t$$
(6)

The results are depicted in Table A.3, A.4, A.5 in the Appendix. Greece is again different, with much of the idiosyncratic movement compared to the EA-12 cycle taking place after 2007Q4. Yet the coefficient remains the lowest of all the peripheral countries. In general, our results support the notion that there is not a striking difference between core and peripheral countries in their reaction to the common EA-12 cycle.<sup>20</sup> However, the goodness of fit for the core regressions as shown by the  $R^2$  is, with the exception of Italy, always lower for the periphery than for the core countries. These findings are broadly in line with Gogas (2013), but at odds with De Grauwe and Ji (2016).



Figure 8: Beta vs. standard deviation

Notes: The scatter plots show the standard deviation of the output gap from Table 5 against the "beta" from regressions in Table 6, 7, 8 (period 1999Q1 - 2015Q4) and Table A.3, A.4, A.5 in the Appendix for the period from 1999Q1-2007Q4 and from 2008Q1-2015Q4.

<sup>&</sup>lt;sup>20</sup>As before, we provide a benchmark analysis that excludes the reference country in the EA-12 series. The results do not vary significantly, see Table A.6 and A.7 in the Appendix.

The large differences in the elasticity with which different countries react to the common euro area cycle begs the question whether these differences can also explain the large differences in the amplitude of national cycles discussed earlier. Figure 8 therefore shows a scatter diagram of the standard deviations in Table 5 against the beta coefficients reported in Tables 7, 6, 8 for the period from 1999Q1-2015Q1 and Tables A.3, A.4, A.5 in the Appendix for the period from 1999Q1-2007Q4 and from 2008Q1-2015Q4.

There is a rather close correlation between the standard deviation of the national business cycles and the degree to which the national cycle reacts to the common one. The exception is, again, Greece, whose economy seems to have followed a completely different pattern since the outbreak of the crisis. The close correlation between the estimated beta coefficients and the standard deviation of national cycles suggests that the much higher variability observed in some countries was not due to specific policy errors, but to the structure of the economies and their financial sectors.

# 5. Conclusion

Business cycle (de)synchronization plays a pivotal role for monetary policy in a currency union. When business cycle conditions diverge strongly, the stance of the ECB, which has to look only at the average, might not fit any country.

We provide evidence for the (de)synchronization of two distinct clusters within the EMU: the core and the periphery. Correlation coefficients indicate that the synchronization of economic activity between these country clusters fell markedly in the period after the start of the financial crisis (2008Q1-2015Q4) compared to the pre-crisis period (1999Q1-2007Q4). Panel and local polynomial regressions based on a quarterly correlation index by Cerqueira (2013) confirm that peripheral countries became less aligned relative to both the core countries and other economies outside of the EMU in the crisis period. Furthermore, the peripheral countries became also less aligned among themselves, in contrast to the cluster of core countries that did not show any change synchronization between the pre- and the crisis period.

We also find that it is not sufficient to look at correlation patterns if one looks for a potential for desynchronization of business cycles. Countries which share the same business cycle might nevertheless experience quite different cyclical positions, and thus require a different monetary policy stance if the amplitude of the cycle is very different. We find indeed large differences in the amplitude of national cycles and the degree to which the national cycle reacts to the common one. A first conclusion is thus that monetary policy making in the euro area faces two problems: the financial crisis led to a desynchronization of the cycle as only the periphery was affected by high risk premia for its sovereign and the private sector. Moreover, individual countries have cycles of different amplitudes, which implies that even countries which share a common cycle might require a different monetary policy stance.

A high degree of correlation is thus not the only condition to ensure "one-sizefits-all". Similarity in the amplitudes of national cycles constitutes another. Cycles have diverged between the core and the periphery (and within the periphery) since the financial crisis. It remains to be seen to what extent this represents a temporary phenomenon. If it was due to the high risk premia during the peak of the crisis, this desynchronization might have been only temporary. It seems that at any rate, the large changes in correlations after the crisis have caused most research to focus on this phenomenon, neglecting the second issue, namely different amplitudes, which can also lead to problems for the common monetary policy.

# References

- AGUIAR-CONRARIA, L., MARTINS, M. M. F. and SOARES, M. J. (2013). Convergence of the economic sentiment cycles in the Eurozone: A time-frequency analysis. *Journal of Common Market Studies*, **51** (3), 377–398.
- and SOARES, M. J. (2011). Business cycle synchronization and the Euro: A wavelet analysis. *Journal of Macroeconomics*, **33** (3), 477–489.
- ALTAVILLA, C. (2004). Do EMU members share the same business cycle? Journal of Common Market Studies, 42 (5), 869–896.
- ANAGNOSTOU, A., PANTELADIS, I. and TSIAPA, M. (2015). Disentangling different patterns of business cycle synchronicity in the EU regions. *Empirica*, 42 (3), 615–641.
- ANGELONI, I. and DEDOLA, L. (1999). From the ERM to the Euro: New evidence on economic and policy convergence among EU countries. ECB Working Paper No. 4.
- ARTIS, M., KROLZIG, H.-M. and TORO, J. (2004). The European business cycle. Oxford Economic Papers, 56 (1), 1–44.
- ARTIS, M. J. (2003). Reflections on the optimal currency area (OCA) criteria in the light of EMU. International Journal of Finance & Economics, 8 (4), 297– 307.
- and ZHANG, W. (2002). Membership of EMU: A fuzzy clustering analysis of alternative criteria. *Journal of Economic Integration*, pp. 54–79.
- BAYOUMI, T. and EICHENGREEN, B. (1992). Shocking aspects of European monetary unification. *NBER Working Paper No. 3949*.
- BEKIROS, S., NGUYEN, D. K., UDDIN, G. S. and SJÖ, B. (2015). Business cycle (de)synchronization in the aftermath of the global financial crisis: Implications for the Euro area. *Studies in Nonlinear Dynamics & Econometrics*, **19** (5), 609– 624.

- BELKE, A. and HEINE, J. M. (2006). Specialisation patterns and the synchronicity of regional employment cycles in Europe. *International Economics and Economic Policy*, 3 (2), 91–104.
- BIERBAUMER-POLLY, J., HUBER, P. and ROZMAHEL, P. (2016). Regional business-cycle synchronization, sector specialization and EU accession. *Jour*nal of Common Market Studies, 54 (3), 544–568.
- BLANCHARD, O., ERCEG, C. and LINDÉ, J. (2015). Jump starting the Euro area recovery: Would a rise in core fiscal spending help the periphery? NBER Working Paper No. 214226.
- BORDO, M. and HELBLING, T. (2003). Have national business cycles become more synchronized? NBER Working Paper No. 10130.
- BURNS, A. F. and MITCHELL, W. C. (1946). *Measuring business cycles*. New York: Columbia University Press (for NBER).
- CAMACHO, M., PEREZ-QUIROS, G. and SAIZ, L. (2006). Are European business cycles close enough to be just one? *Journal of Economic Dynamics and Control*, **30** (9–10), 1687–1706.
- CAMERON, A. C. and TRIVEDI, P. K. (2005). Microeconometrics: methods and applications. Cambridge University Press.
- CAMPOS, N. F. and MACCHIARELLI, C. (2016). Core and Periphery in the European Monetary Union: Bayoumi and Eichengreen 25 years later. *Economics Letters*, 147, 127–130.
- CANOVA, F. (1998). Detrending and business cycle facts. Journal of Monetary Economics, 41 (3), 475–512.
- CERQUEIRA, P. A. (2013). A closer look at the world business cycle synchronization. International Economics and Economic Policy, 10 (3), 349–363.
- and MARTINS, R. (2009). Measuring the determinants of business cycle synchronization using a panel approach. *Economics Letters*, **102** (2), 106–108.

- CESARONI, T. and DE SANTIS, R. (2016). Current Account 'Core-Periphery Dualism' in the EMU. *forthcoming, The World Economy.*
- CHRISTODOULAKIS, N., DIMELIS, S. P. and KOLLINTZAS, T. (1995). Comparisons of business cycles in the EC: Idiosyncracies and regularities. *Economica*, pp. 1–27.
- CLARIDA, R., GALÍ, J. and GERTLER, M. (1999). The science of monetary policy: A new keynesian perspective. *Journal of Economic Literature*, 37, 1661–1707.
- CROUX, C., FORNI, M. and REICHLIN, L. (2001). A measure of comovement for economic variables: Theory and empirics. *Review of Economics and Statistics*, 83 (2), 232–241.
- DARVAS, Z. and SZAPÁRY, G. (2008). Business cycle synchronization in the enlarged EU. Open Economies Review, 19 (1), 1–19.
- DE GRAUWE, P. and JI, Y. (2016). Flexibility versus Stability. A difficult tradeoff in the Eurozone. *CEPR Discussion Paper 11372*.
- DEGIANNAKIS, S., DUFFY, D. and FILIS, G. (2014). Business cycle synchronization in EU: A time-varying approach. *Scottish Journal of Political Economy*, 61 (4), 348–370.
- DICKERSON, A. P., GIBSON, H. D. and TSAKALOTOS, E. (1998). Business cycle correspondence in the European Union. *Empirica*, 25 (1), 49–75.
- EICHENGREEN, B. (1990). One money for Europe? Lessons from the US currency union. *Economic Policy*, 5 (10), 117–187.
- (1991). Is Europe an optimum currency area? NBER Working Paper No. 3579.
- (1993). European monetary unification. Journal of Economic Literature, **31** (3), 1321–1357.
- EUROPEAN COMMISSION (1990). One Market, one money An evaluation of the potential benefits and costs of forming an economic and monetary union. *European Economy*, (44).

- EUROPEAN COMMISSION (2014). Focus: growth differences between EA member states since the crisis. *Quarterly Report on the Euro Area*, (13), 7–20.
- FAN, J. and GIJBELS, I. (1996). Local polynomial modelling and its applications. London: Chapman Hall.
- FATAS, A. (1997). EMU: Countries or regions? Lessons from the EMS experience. European Economic Review, 41 (3), 743–751.
- FELDSTEIN, M. (1997). The political economy of the European Economic and Monetary Union: Political sources of an economic liability. *Journal of Economic Perspectives*, **11** (4), 23–42.
- FERREIRA-LOPES, A. and PINA, Á. M. (2011). Business cycles, core, and periphery in monetary unions: Comparing Europe and North America. Open Economies Review, 22 (4), 565–592.
- FERRONI, F. and KLAUS, B. (2015). Euro area business cycles in turbulent times: Convergence or decoupling? Applied Economics, 47 (34-35), 3791–3815.
- FIDRMUC, J. and KORHONEN, I. (2010). The impact of the global financial crisis on business cycles in Asian emerging economies. *Journal of Asian Economics*, 21 (3), 293–303.
- FRANKEL, J. A. and ROSE, A. K. (1997). Is EMU more justifiable ex post than ex ante? *European Economic Review*, 41 (3), 753–760.
- and (1998). The endogeneity of the optimum currency area criteria. *Economic Journal*, **108** (449), 1009–1025.
- FURCERI, D. and KARRAS, G. (2008). Business-cycle synchronization in the EMU. Applied Economics, 40 (12), 1491–1501.
- GÄCHTER, M. and RIEDL, A. (2014). One money, one cycle? The EMU experience. Journal of Macroeconomics, 42, 141–155.
- —, RIEDL, A. and RITZBERGER-GRÜNWALD, D. (2012). Business cycle synchronization in the Euro area and the impact of the financial crisis. *Monetary Policy* & the Economy, 2, 33–60.

- GAYER, C. (2007). A fresh look at business cycle synchronisation in the Euro area. European Economy Economic Papers, (287).
- GOGAS, P. (2013). Business cycle synchronisation in the European Union. Journal of Business Cycle Measurement and Analysis, 2013 (1), 1–14.
- GONÇALVES, C. E. S., RODRIGUES, M. and SOARES, T. (2009). Correlation of business cycles in the Euro zone. *Economics Letters*, **102** (1), 56–58.
- GOUVEIA, S. and CORREIA, L. (2008). Business cycle synchronisation in the Euro area: the case of small countries. *International Economics and Economic Policy*, 5 (1-2), 103–121.
- GRIGORAŞ, V. and STANCIU, I. E. (2016). New evidence on the (de)synchronisation of business cycles: Reshaping the European business cycle. *International Economics*, 147, 27–52.
- GROS, D. and HEFEKER, C. (2002). One size must fit all: National divergences in a monetary union. *German Economic Review*, **3** (3), 247–262.
- HAAN, J. D., INKLAAR, R. and JONG-A-PIN, R. (2008). Will business cycles in the Euro area converge? A critical survey of empirical research. *Journal of economic surveys*, **22** (2), 234–273.
- HARDING, D. and PAGAN, A. (2002). Dissecting the cycle: A methodological investigation. Journal of Monetary Economics, 49 (2), 365–381.
- and (2006). Synchronization of cycles. Journal of Econometrics, 132 (1), 59–79.
- HODRICK, R. J. and PRESCOTT, E. C. (1997). Postwar US business cycles: An empirical investigation. *Journal of Money, Credit, and Banking*, pp. 1–16.
- IMBS, J. (2010). The first global recession in decades. *IMF economic review*, 58 (2), 327–354.
- INKLAAR, R., JONG-A-PIN, R. and HAAN, J. D. (2008). Trade and business cycle synchronization in OECD countries - a re-examination. *European Economic Review*, **52** (4), 646–666.

- KONSTANTAKOPOULOU, I. and TSIONAS, E. (2011). The business cycle in Eurozone economies (1960 to 2009). Applied Financial Economics, 21 (20), 1495–1513.
- KOSE, M. A., OTROK, C. and WHITEMAN, C. H. (2003). International business cycles: World, region, and country-specific factors. *American Economic Review*, pp. 1216–1239.
- LANE, P. R. (2012). The European sovereign debt crisis. Journal of Economic Perspectives, 26 (3), 49–68.
- LEE, J. (2013). Business cycle synchronization in Europe: Evidence from a dynamic factor model. *International Economic Journal*, 27 (3), 347–364.
- LEHWALD, S. (2013). Has the Euro changed business cycle synchronization? Evidence from the core and the periphery. *Empirica*, 40 (4), 655–684.
- LUCAS, R. E. (1977). Understanding business cycles. Carnegie-Rochester conference series on public policy, 5, 7–29.
- MCKINNON, R. I. (1963). Optimum currency areas. American Economic Review, 53 (4), 717–725.
- MONGELLI, F. P. (2008). European economic and monetary integration, and the optimum currency area theory. *European Economy Economic Papers*, **302**.
- MUNDELL, R. A. (1961). A theory of optimum currency areas. American Economic Review, 51 (4), 657–665.
- PAPAGEORGIOU, T., MICHAELIDES, P. G. and MILIOS, J. G. (2010). Business cycles synchronization and clustering in Europe (1960–2009). *Journal of Economics and Business*, **62** (5), 419–470.
- RACINE, J. (1997). Consistent significance testing for nonparametric regression. Journal of Business & Economic Statistics, 15 (3), 369–378.
- —, HART, J. and LI, Q. (2006). Testing the significance of categorical predictor variables in nonparametric regression models. *Econometric Reviews*, **25** (4), 523– 544.

- RAVN, M. O. and UHLIG, H. (2002). On adjusting the Hodrick-Prescott filter for the frequency of observations. *Review of Economics and Statistics*, 84 (2), 371–376.
- ROGOFF, K. (1985). The optimal degree of commitment to an intermediate monetary target. *The Quarterly Journal of Economics*, pp. 1169–1189.
- SCHÄFER, D. (2016). A banking union of ideas? The impact of ordoliberalism and the vicious circle on the EU banking union. *Journal of Common Market Studies*, 54 (4), 961–980.
- WEYERSTRASS, K., VAN AARLE, B., KAPPLER, M. and SEYMEN, A. (2011). Business cycle synchronisation with (in) the Euro area: In search of a 'Euro effect'. *Open Economies Review*, **22** (3), 427–446.
- WILLETT, T. D., PERMPOON, O. and WIHLBORG, C. (2010). Endogenous OCA analysis and the early Euro experience. *The World Economy*, **33** (7), 851–872.
- WORTMANN, M. and STAHL, M. (2016). One size fits some: A reassessment of EMU's core–periphery framework. *Journal of Economic Integration*, **31** (2), 377–413.

# A. Appendix

	gre	ita	$_{\rm spa}$	ire	$\operatorname{por}$
EA-11	0.377	1.009***	$0.794^{***}$	1.621***	$0.738^{***}$
	(0.481)	(0.0626)	(0.156)	(0.197)	(0.139)
$R^2$	0.040	0.868	0.561	0.651	0.452

Table A.1: Contemporaneous effect of EA-11 cycle on national cycle: periphery

Notes: Dependent variable is the quarterly (HP filtered) cyclical component of each peripheral country against the EA-12 cycle excluding the reference country. Sample from 1999Q1 - 2015Q4. HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

Table A.2: Contemporaneous effect of EA-11 cycle on national cycle: core

	Finland	France	Germany	Austria	Netherlands
EA-11	$1.505^{***}$	$0.675^{***}$	$1.270^{***}$	$0.942^{***}$	$1.026^{***}$
	(0.155)	(0.0052)	(0.158)	(0.0574)	(0.102)
$R^2$	0.835	0.818	0.782	0.831	0.819

<u>Notes</u>: Dependent variable is the quarterly (HP filtered) cyclical component of each core country, independent variable is the EA-12 cycle excluding the reference country. Sample from 1999Q1 - 2015Q4. HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

	Greece	Italy	Spain	Ireland	Portugal
EA-12	$\begin{array}{c} 0.734^{**} \\ (0.342) \end{array}$	$\begin{array}{c} 0.936^{***} \\ (0.0625) \end{array}$	$\begin{array}{c} 0.883^{***} \\ (0.0627) \end{array}$	$\begin{array}{c} 1.819^{***} \\ (0.165) \end{array}$	$\begin{array}{c} 0.812^{***} \\ (0.134) \end{array}$
Crisis	$\begin{array}{c} 0.00341 \\ (0.0120) \end{array}$	-0.000244 (0.00181)	0.000687 (0.00407)	-0.00699 (0.00505)	0.00139 (0.00490)
Crisis $\times$ EA-12	-0.429 (0.725)	$0.169^{**}$ (0.0847)	-0.0113 (0.239)	-0.418 (0.299)	-0.0827 (0.228)
$R^2$	0.078	0.918	0.651	0.701	0.475

Table A.3: Contemporaneous effect of EA-12 cycle on national cycle: periphery

Notes: Dependent variable is the quarterly (HP filtered) cyclical component of each peripheral country against the EA-12 with a sample from in 1990Q1 and ends in 2015Q4. The crisis dummy covers the period from "2008Q1 - 2015Q4". HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

Table A.4: Contemporaneous effect of EA-12 cycle on national cycle: core

	Finland	France	Germany	Austria	Netherlands
EA-12	$\begin{array}{c} 1.317^{***} \\ (0.169) \end{array}$	$\begin{array}{c} 0.773^{***} \\ (0.0736) \end{array}$	$\frac{1.215^{***}}{(0.0764)}$	$\begin{array}{c} 0.964^{***} \\ (0.0504) \end{array}$	$\frac{1.183^{***}}{(0.142)}$
Crisis	$\begin{array}{c} -0.000346 \\ (0.00302) \end{array}$	$\begin{array}{c} -0.00190\\ (0.00171) \end{array}$	$\begin{array}{c} 0.000859 \\ (0.00249) \end{array}$	$\begin{array}{c} 0.000244 \\ (0.00214) \end{array}$	0.00163 (0.00238)
Crisis $\times$ EA-12	$0.325 \\ (0.223)$	-0.0765 (0.102)	0.0787 (0.152)	-0.0242 (0.113)	-0.246 (0.164)
$R^2$	0.853	0.879	0.906	0.840	0.855

<u>Notes</u>: Dependent variable is the quarterly (HP filtered) cyclical component of each core country against the EA-12 cycle with a sample from 1999Q1 to 2015Q4. The crisis dummy covers the period from "2008Q1 - 2015Q4". HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

	Sweden	Denmark	Norway	Switzerland
EA-12	$\begin{array}{c} 1.119^{***} \\ (0.137) \end{array}$	$\begin{array}{c} 1.046^{***} \\ (0.144) \end{array}$	$\begin{array}{c} 0.674^{***} \\ (0.190) \end{array}$	$\frac{1.151^{***}}{(0.119)}$
Crisis	-0.00393 (0.00325)	-0.00186 (0.00303)	-0.000271 (0.00329)	$\begin{array}{c} 0.00212 \\ (0.00213) \end{array}$
Crisis $\times$ EA-12	0.244 (0.263)	-0.0598 (0.171)	-0.400** (0.190)	$-0.474^{**}$ (0.192)
$R^2$	0.825	0.762	0.326	0.849

Table A.5: Contemporaneous effect of EA-12 cycle on national cycle: outside EMU

Notes: Dependent variable is the quarterly (HP filtered) cyclical component of each non-EMU country against the EA-12 cycle with a sample from 1999Q1 to 2015Q4. The crisis dummy covers the period from "2008Q1 - 2015Q4". HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

Table A.6: Contemporaneous effect of EA-11 cycle on national cycle: periphery

	Greece	Italy	Spain	Ireland	Portugal
EA-11	$0.698^{*}$ (0.350)	$\begin{array}{c} 0.891^{***} \\ (0.0690) \end{array}$	$\begin{array}{c} 0.863^{***} \\ (0.0700) \end{array}$	$\begin{array}{c} 1.824^{***} \\ (0.174) \end{array}$	$\begin{array}{c} 0.797^{***} \\ (0.135) \end{array}$
Crisis	$\begin{array}{c} 0.00320 \\ (0.0121) \end{array}$	$\begin{array}{c} -0.000408 \\ (0.00219) \end{array}$	$\begin{array}{c} 0.000630 \\ (0.00460) \end{array}$	$\begin{array}{c} -0.00725 \\ (0.00519) \end{array}$	$\begin{array}{c} 0.00136 \\ (0.00499) \end{array}$
Crisis $\times$ EA-11	-0.541 (0.713)	$0.216^{**}$ (0.100)	-0.117 (0.223)	-0.436 (0.303)	-0.0923 (0.225)
$R^2$	0.063	0.877	0.565	0.684	0.456

<u>Notes</u>: Dependent variable is the quarterly (HP filtered) cyclical component of each peripheral country against the EA-12 cycle excluding the respective country. The crisis dummy covers the period from "2008Q1 - 2015Q4". HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.

Table A.7: Contemporaneous effect of EA-11 cycle on national cycle: core

	Finland	France	Germany	Austria	Netherlands
EA-11	$\begin{array}{c} 1.315^{***} \\ (0.175) \end{array}$	$\begin{array}{c} 0.716^{***} \\ (0.0778) \end{array}$	$\begin{array}{c} 1.245^{***} \\ (0.124) \end{array}$	$\begin{array}{c} 0.959^{***} \\ (0.0520) \end{array}$	$ \begin{array}{c} 1.177^{***} \\ (0.156) \end{array} $
Crisis	$\begin{array}{c} -0.000382 \\ (0.00309) \end{array}$	-0.00229 (0.00199)	$\begin{array}{c} 0.000918 \\ (0.00374) \end{array}$	$\begin{array}{c} 0.000233 \\ (0.00221) \end{array}$	0.00167 (0.00258)
Crisis $\times$ EA-11	$\begin{array}{c} 0.340 \\ (0.231) \end{array}$	-0.0918 (0.117)	$0.0562 \\ (0.287)$	-0.0284 (0.118)	-0.253 (0.178)
$R^2$	0.845	0.835	0.783	0.831	0.833

Notes: Dependent variable is the quarterly (HP filtered) cyclical component of each core country against the EA-11 cycle excluding the respective country. The crisis dummy covers the period from "2008Q1 - 2015Q4". HAC standard errors in parenthesis. \*\*\*/\*\*/\* indicate significance at the 1%/5%/10% level.